Author’s response to reviews

Title: The multiple systemic artery to pulmonary artery fistulas resulting in severe irreversible pulmonary arterial hypertension in patient with previous history of pneumothorax.

Authors:

Wojciech Jacheć (wjachec@interia.pl)
Andrzej Tomasik (tomask@poczta.onet.pl)
Marcin Kurzyna (marcin.kurzyna@ecz-otwoc.pl)
Radosław Pietura (radoslawpietur@poczta.onet.pl)
Adam Torbicki (adam.torbicki@ecz-otwoc.pl)
Jan Głowacki (j.glowacki@sccs.pl)
Ewa Nowalany-Kozielska (ewakozielska@wp.pl)
Celina Wojciechowska (wojciechowskac@wp.pl)

Version: 1 Date: 13 Oct 2018

Author’s response to reviews:

Dear Editor

We would like to inform you that we have written informed consent from wife of patient to use her husband's medical data to prepare a case report.

In addition, according to the recommendations, we have removed figure titles and capitations from the figure files and placed its in Figure legend subsection

Sincerely

In the name of co-authors

Wojciech Jacheć